Author’s response to reviews

Title: "I prefer a male nurse to a female nurse": Patients' preference for, and satisfaction with nursing care provided by male nurses at the Komfo Anokye Teaching Hospital.

Authors:

Hayford Budu (buduhayford@yahoo.com)
Emmanuel Abalo (eabalo92@gmail.com)
Victoria Bam (elormbam@yahoo.com)
Deus Agyemang (deusoseiagyemang@gmail.com)
Shirley Noi (queenelizabert5@gmail.com)
Florence Budu (florencea890@gmail.com)
Prince Peprah (princepeprah15@gmail.com)

Version: 2 Date: 30 Aug 2018

Author’s response to reviews:

We have had the opportunity to address the comments provided by the reviewers. Justification for certain statements, and corrections of others, have been included in the revised manuscript. Also, a point-to-point response to the concerns raised by the reviewers has been attached for their perusal. On behalf of my co-authors, we express our profound gratitude for the opportunity granted us to revise and resubmit our manuscript. We hope the revision meets the standards of the journal and fulfills the expectation of the reviewers, and do look forward to the publication of this article.

Thank you.